Author’s response to reviews

Title: Effects of Depression, Dementia and Delirium on Activities of Daily Living in Elderly Patients after Discharge

Authors:
Ching Fu Weng (philweng11@yahoo.com.tw)
Kun-Pei Lin (dtmed512@gmail.com)
Feng-Ping Lu (lufp1996@gmail.com)
Jen-Hau Chen (jhhchen@ntu.edu.tw)
Chiung-Jung Wen (cindy65111@gmail.com)
Jui-Hua Peng (pengjh@ntu.edu.tw)
Ailun Tseng (heather_tseng@hotmail.com)
Ding-Cheng Chan (doctord6226@yahoo.com)

Version: 3 Date: 24 Sep 2019

Author’s response to reviews:

Dear Editor,

We really appreciate for all your help during submission. The followings are the reply to your comments.

1. Thank you for providing information regarding consent obtained from participants. In the Methods section and Ethics approval section of the Declarations, please state whether verbal consent was approved by the ethics committee. Please also provide additional details about proxy consent, including examples of proxy relationships to participants.

Reply.

Thank you for your reminder. We did not describe the process of informed consent clearly in the original paragraph. The patients or their proxies were informed verbally about the present study and provided with the written consent form. After deliberate consideration, they could make a decision whether to participate in the study or not. All the participants enrolled in the study signed the informed consent form by himself/ herself or their proxies.
2. In the Authors’ contributions section, please represent the authors using their initials only, instead of any full names. Please also include a statement in Authors’ contributions indicating that all authors read and approved the final version of the manuscript.

Reply.

Thank you for your kindly reminder. We have revised the contributions section.

3. In your Funding section, please also state the role of the funding body in the design of the study; collection, analysis, and interpretation of data; and in writing the manuscript.

Reply.

Thank you for your suggestion. We have added the role of the funding resource in the Funding section.

4. Please ensure Figure 1 is referenced in the main text of the manuscript. Any items which are not cited may be deleted by our production department upon publication.

Reply.

Thank you for your kindly reminder. We have put the citation of Figure 1 in the main text.

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colors. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Reply.

Thanks again for your reminder!